

<mark>\_</mark>



# **Geophysical Research Letters**<sup>•</sup>

# **RESEARCH LETTER**

10.1029/2025GL114807

#### **Key Points:**

- The water surface slope of the Yangtze near the confluence can be halved due to backwater effects
- The backwater effects can extend to around 74 km upstream of the confluence
- The highest elevated water level at Jiujiang (24 km upstream of the confluence) is estimated to be up to 14.1 m

#### **Supporting Information:**

Supporting Information may be found in the online version of this article.

#### Correspondence to:

L. Jiang, jianglg@sustech.edu.cn

#### Citation:

Meng, Y., Jiang, L., Du, E., Zhang, X., Wang, W., & Wang, L. (2025). A new understanding of the Poyang Lake-Yangtze River interaction: A backwater effect on the Yangtze River perspective. *Geophysical Research Letters*, *52*, e2025GL114807. https://doi. org/10.1029/2025GL114807

Received 10 JAN 2025 Accepted 18 MAR 2025

#### **Author Contributions:**

Conceptualization: Liguang Jiang Data curation: Liguang Jiang, Erhu Du, Xingxing Zhang, Lei Wang Formal analysis: Yuanyuan Meng, Liguang Jiang Funding acquisition: Liguang Jiang Methodology: Yuanyuan Meng, Liguang Jiang Project administration: Liguang Jiang Software: Yuanyuan Meng Supervision: Liguang Jiang Validation: Yuanyuan Meng Visualization: Yuanyuan Meng Writing – original draft: Yuanyuan Meng, Liguang Jiang

© 2025 The Author(s). This is an open access article under the terms of the Creative Commons Attribution-NonCommercial License, which permits use, distribution and reproduction in any medium, provided the original work is properly cited and is not used for commercial purposes.

# A New Understanding of the Poyang Lake-Yangtze River Interaction: A Backwater Effect on the Yangtze River Perspective

# Yuanyuan Meng<sup>1</sup>, Liguang Jiang<sup>1,2</sup>, Erhu Du<sup>3</sup>, Xingxing Zhang<sup>4</sup>, Weijie Wang<sup>5</sup>, and Lei Wang<sup>6</sup>

<sup>1</sup>School of Environmental Science and Engineering, Southern University of Science and Technology, Shenzhen, China, <sup>2</sup>Changjiang River Scientific Research Institute (CRSRI), Changjiang Water Resources Commission, Wuhan, China, <sup>3</sup>Yangtze Institute for Conservation and Development, Hohai University, Nanjing, China, <sup>4</sup>State Key Laboratory of Resources and Environmental Information System, Institute of Geographic Sciences and Natural Resources Research, Chinese Academy of Sciences, Beijing, China, <sup>5</sup>State Key Laboratory of Simulation and Regulation of Water Cycle in River Basin, China Institute of Water Resources and Hydropower Research, Beijing, China, <sup>6</sup>Henan Key Laboratory of Water Resources Conservation and Intensive Utilization in the Yellow River Basin, North China University of Water Resources and Electric Power, Zhengzhou, China

**Abstract** Interconnected river-lake systems are much more complex than either one due to their interactions. The Poyang Lake-Yangtze River system has received much attention due to the reverse flow from the Yangtze River, which blocks the flow of Poyang Lake and causes extensive inundation. However, a less explored perspective is the backwater effects of lake outflow on the Yangtze River. This study seeks to fill this gap by investigating the water surface slope (WSS) along the 294 km reach upstream of the confluence. Results show that huge variations of WSS (the maximum doubles the minimum) occur due to the dynamics of lake outflow and river discharge. The backwater can extend upstream to around 74 km. Elevated water level at Jiujiang can be up to 14.1 m in the worst case. The findings can advance our understanding of complex hydrodynamic processes in river-lake systems and provide important policy implications for flood management.

**Plain Language Summary** River networks are very complex. When two rivers meet, they can lead to complicated hydraulic characteristics at the confluence. For instance, their interaction can cause variations in flow velocity by giving an additional gradient in water level, which is known as a backwater effect. Such an effect can cause severe flood events due to the elevated water levels. In this study, we investigated the effects of Poyang Lake outflow on the Yangtze River from a novel perspective of river surface slope. With the advancements in wide-swath altimetry, the hydraulic characteristics of river-lake confluence can be better quantified. The findings shown in this study are essential for flood control, flood risk evaluation, and engineering issues.

# 1. Introduction

Rivers and lakes are fundamental landscape units, and supply water that supports many forms of life (Poff, 2019). Connected rivers and lakes, often referred to as river-lake systems, are more complex than either due to the river-lake interactions (Lesack & Marsh, 2010). The Yangtze River-Poyang Lake system in China is such a complex system (Hu et al., 2007). However, instead of river-lake connection in series, Poyang Lake is a major tributary recharging the Yangtze River. There are five major rivers draining a total area of about 136,000 km<sup>2</sup> into the Poyang Lake, and then the lake naturally discharges an average of about 140 km<sup>3</sup> per year to the Yangtze River. Poyang Lake is a critical habitat for fish species and many migratory birds. However, the hydrological regime has undergone pronounced changes during the past two decades, and the ecosystem is under threat from both human activities and climate change (Q. Zhang et al., 2023). The dynamic lake-river interactions have received much attention in the context of recent extreme events. Most studies focus on the exchange of water, sediment, and pollutants between lakes and rivers, as well as hydrodynamic modeling in the context of climate change and human activities (M. Chen et al., 2022; Guo et al., 2012; Hu et al., 2007; Yang et al., 2016). In the recent decade, more effort has been put into understanding the severe and extended dry periods of Poyang Lake (Feng et al., 2013; Guo et al., 2012; Wang et al., 2017; Yang et al., 2016). For instance, Li et al. (2023)



Writing – review & editing: Liguang Jiang, Erhu Du, Xingxing Zhang, Weijie Wang, Lei Wang found that hydrological droughts became increasingly severe from 1964 to 2016, and such changes may relate to large-scale climate systems.

However, previous studies mostly focus on the impacts of Yangtze River on the Poyang Lake. A less explored perspective is the backwater effects of Poyang Lake on the Yangtze River. Broadly, the backwater effects may occur when one or more downstream elements, such as lakes, tributaries, ponds, dams, levees, and tides, exist. Consequently, the river flow will be blocked, and water will be backed up in its course by the downstream elements such that the water level is higher than the smallest level required for conveying the same discharge (Chow, 1959; Petersen-Øverleir & Reitan, 2009). Therefore, the WSS gets smaller due to the loss of water energy for a similar discharge over the affected river reach (Castelltort et al., 2020; Hidayat et al., 2011). One of the most representative examples of such backwater effects is the Amazon river system (Meade et al., 1991; Trigg et al., 2009), where the lengths of reach affected by backwater were estimated to be at least 300 km, 810 km, and 890 km in the Negro River, the Madeira River, and the Purus River, respectively. Moreover, the elevated river level can be 2–3 m at a station even 260 km upstream of the Madeira River, which leads to a halved river surface slope (Meade et al., 1991). The backwater effects at confluences of many other rivers have also been studied, such as the Mississippi River (Edwards et al., 2016; Lamb et al., 2012), the Missouri River (Liu et al., 2023), and so on. Practically, this elevated water level can increase the flood risk vulnerability in flood-prone areas (Jiang et al., 2020; Luo et al., 2017; X. Zhang et al., 2023).

The backwater effects of lake outflow are widely reported. For instance, the backwater flow from lakes (e.g., Lake Tana, Lake Kyoga, Lake Victoria, etc.) in the Nile River basin has been widely studied (Sutcliffe & Brown, 2018; TATE et al., 2001). Abate et al. (2015) attributed the sediment deposition in the Gumara River of the upper Blue Nile basin to the backwater effect of Lake Tana. Petersen-Øverleir and Reitan (2009) found that Lake Vansjø backed up the flow in the Guthusbekken stream for almost a month during a major flood event. Moreover, Hidayat et al. (2011) concluded that the backwater effects from lakes and tributaries have a considerable impact on river discharge estimates. Lake outflow changes the flow regime not only upstream (backwater) but also downstream (drawdown).

Poyang Lake and Yangtze River are more complex than other run-of-river lake-river systems. However, it has not been well studied in the literature that how Poyang Lake affects Yangtze River. Several questions remain to be answered. Is the backwater effect of the Poyang Lake negligible or pronounced? To what extent does the backwater affect the Yangtze River? How does the interplay of the Poyang Lake outflow and the Yangtze River discharge affect the backwater events? Therefore, the overarching goal of this study is to investigate and analyze the response of Yangtze River flow to the outflow of Poyang Lake from the perspective of water level change and associated river WSS.

# 2. Data and Methods

#### 2.1. Study Area

The Yangtze-Poyang Lake system is one of the largest river-lake systems worldwide. In this study, we focus on the reach between Hankou station (GS\_1) and Pengze station (GS\_6), which is about 335 km. As shown in Figure 1, the studied reach is separated into small reaches by the six gauging stations, which are numbered in downstream order. The Poyang Lake joins the Yangtze River around the Hukou station (GS\_5), where the elevation difference between GS\_5 and the confluence is negligible (see Text S1 and Table S1 in Supporting Information S1). The Jiujiang-Hukou reach (RR\_4-5) and the Hukou-Pengze reach (RR\_5-6) are the adjacent upstream and downstream reaches referencing the Hukou station. In general, the slope of the whole reach is gentle as shown in Figure 1. Clearly, DEM-derived water slopes of RR\_2-3 and RR\_3-4 are negative most likely due to lower accuracy of SRTM DEM. Therefore, we estimate the slopes based on dry season water levels.

### 2.2. Data

The main gauge data used are water level and discharge from the Hubei Hydroinfo website. Daily water levels between 2007 and 2022 were collected at all six gauging stations, while discharge was only available at GS\_1, GS\_4, and GS\_5 for the same period. Below, we used the symbols Q1, Q4, and Q5 to indicate the discharge at GS\_1, GS\_4, and GS\_5, respectively. Q5 represents the Poyang Lake outflow. In principle, Water Surface Slope (WSS) can be calculated for reaches between any two stations, which is indicated by the acronym WSS followed





**Figure 1.** A detailed river network of the Yangtze River between Hankou station and Pengze station, along with the six gauging stations. Major tributaries of the Yangtze River are shown with labeled names. Rivers and lake were obtained from National Geomatics Center of China (https://www.ngcc.cn/). The left inset plot shows the river longitudinal profile, with the chainage referencing to Hankou station. Note that the insitu data in the profile is the median of observed water level during the dry season (Dec, Jan, and Feb) over the period from 2007 to 2020 while DEM from MERIT DEM.

by the station numbers (e.g., WSS\_1-4 for the slope of the reach between stations GS\_1 and GS\_4). In addition, river cross-sectional profiles at station GS\_4 were also collected from the China Hydrological Yearbook to calculate the water depth.

#### 2.3. Methods

Based on the twin-gauge approach, the WSS of each river reach can be calculated using the equation below:

WSS = 
$$\frac{\Delta Z}{x}$$
 (1)

where x is the reach length (m),  $\Delta Z$  is the water level difference at the twin gauges of the reach of interest (m). The WSS values are expressed in units of cm/km.

To understand the raised water level by backwater, the difference between the normal depth and actual water depth is used and calculated as follows:

$$y_{\rm diff} = y_a - y_n \tag{2}$$

where  $y_n$  is the normal water depth, which refers to the water depth under steady uniform flow, and  $y_a$  is the actual water depth.  $y_a$  can be directly calculated by subtracting the channel datum from the water level.

Below, we describe the way to calculate  $y_n$ . Since we only have the cross-section profiles at station GS\_4, the  $y_a$  is thus only calculated for this section. Here, we assume the riverbed gradient is constant between GS\_4 and GS\_5.

Based on surveyed cross-section data, the river width (w), wetted perimeter (P), and flow area (A) corresponding to the daily water level can be calculated. Thus, we can obtain the hydraulic radius R, that is, A/P. Therefore, we can express the discharge using Manning's equation as:



$$Q = \frac{1}{n} \mathrm{AR}^{2/3} S_f^{1/2}$$
(3)

where Q donates the discharge,  $S_f$  is the friction slope, which can be approximated by the riverbed slope ( $S_0$ ) under steady flow. Here,  $S_f$  is calculated based on the water levels in November 2022 (Text S2 and Figure S1 in Supporting Information S1) when we assume the flow is in a steady state. Note that the choice of dry season months has marginal impact on the calculation of  $S_f$ , which consequently has little influence on  $y_{diff}$  (refer to Table S2 in Supporting Information S1). In this way, we can resolve the Manning's n as below:

$$n = \frac{1}{Q} A R^{2/3} S_0^{1/2}$$
 (4)

Based on the data during the dry season (Nov to Feb, see Text S3, Figure S2b in Supporting Information S1), the calculated range of n is between 0.026 and 0.053. The accurate Manning's n for a certain water level is difficult to determine. It is well known that Manning's n decreases with the increase of discharge or depth for large rivers (Alves et al., 2020; Tahmid et al., 2021). Here, we assume that n is inversely proportional to the water level, as expressed below:

$$n = n_{\max} - (n_{\max} - n_{\min}) \times (y_a / y_a^{\max})$$
<sup>(5)</sup>

Linear instead of exponential relationships used is because of the lack of data to fit such a function. Since Manning's *n* can be slightly different in the wet seasons, different values of the maximum  $(n_{max})$  and minimum  $(n_{min})$  were used to consider the uncertainty. Based on the values calculated using dry season flows, the maximum  $(n_{max})$  and minimum  $(n_{min})$  are in the range of [0.05, 0.06] and [0.005, 0.03], respectively, and with an interval of 0.005. Therefore, a total of 18 combinations of  $n_{min}$  and  $n_{max}$  were obtained. Consequently, the calculated normal depth  $y_n$  and raised water level  $y_{diff}$  also have 18 values.

Given that  $y_n$  is a function of cross-sectional geometry (i.e., AR<sup>2/3</sup>, see Figure S3 in Supporting Information S1), we just solve Equation 3 for AR<sup>2/3</sup> under the steady uniform flow condition to determine  $y_n$ . Substituting the values of above variables under steady uniform state, we can get the normal water depth  $y_n$ :

$$y_n = \left(\frac{Q \cdot n}{\alpha \cdot S_f^{\frac{1}{2}}}\right)^{\frac{1}{p}} \tag{6}$$

where  $\alpha$  and  $\theta$  are the fitting parameters derived from the relationship AR<sup>2/3</sup> ~  $y_n$ . Considering the varying nature of the cross-sectional profile, we fit a power-law function between  $y_n$  and AR<sup>2/3</sup> for each year from 2011 to 2020 (Table S3 in Supporting Information S1).

#### 3. Results

#### 3.1. Backwater Effects at the Confluence

Figure 2a shows the daily WSS variations of 11 reaches over a long term. It is obvious that the WSS for most reaches less variable throughout the year, while WSS\_4-5 is remarkably higher during the dry seasons than wet seasons. Nevertheless, the Froude number (Fr) at GS\_4 throughout the year is always far less than 1, indicating that the flow is always in subcritical state, excluding the flow state change (Fr < 0.2, see Text S4 and Figure S4 in Supporting Information S1). Clearly, close to the river-lake confluence, WSS\_4-5 and WSS\_5-6 have very distinct temporal patterns. Specifically, WSS\_4-5 quickly declines at mid-February. WSS\_4-5 reaches three local minimums at the end of March, June, and September, respectively, and reaches the maximum at the end of March, reaching a maximum in June and July, and resuming the lower value at the end of December. Moreover, WSS\_5-6 even exceeds WSS\_4-5 from June to July. Such a phenomenon is mainly due to the backwater effects because, in a normal state, the WSS of a river reach does not change dramatically, not to mention over such a long period as 12 months in a multi-year average condition.



**Figure 2.** Intra-annual variation of water surface slope (WSS) at the multi-year (2007–2020) level for the total of 11 river reaches from Hankou to Pengze. (a) The line indicates the median daily WSS value for each river reach, and the shaded area represents the 5%–95% confidence interval. (b) The boxplot of all daily WSS values for the 11 river reaches over the entire period, with the gray line connecting the median values of each group. (c) Scatter plots of the relationships between WSS\_4–5 and the corresponding Q5 and Q5/Q1. (d) Similar to (c), but for WSS\_5–6.

This pattern is very well aligned with the ratio of lake outflow and river flow (Q5/Q1, see Figure S5 in Supporting Information S1). As the ratio increases, the backwater effects are more pronounced, leading to a raised water level at the confluence. Consequently, WSS\_4-5 declines and WSS\_5-6 increases, respectively. However, it has to be noted that the backwater effects not only rely on the flow ratio but also the absolute value of the lake outflow (Q5), as illustrated in Figures 2c and 2d. For instance, when the ratio is about 0.3, the backwater would be stronger when Q5 is larger, that is, larger WSS\_5-6 (darker triangles) and smaller WSS\_4-5 (lighter circles).

The intra-annual variability of WSS is pronounced, especially for the reaches close to the confluence, as indicated by the thick lines and boxplots in Figures 2a and 2b, respectively. The WSS is nearly halved or doubled for WSS\_4-5 and WSS\_5-6 in June and July when the flows are larger (Figure S5b in Supporting Information S1), and the backwater effects are also stronger. Throughout the year, the long-term daily median value of WSS\_4-5 varies between 1.77 and 3.29 cm/km, while the median value of WSS\_5-6 varies between 1.18 and 2.24 cm/km. Moreover, the WSS values of different river reaches are markedly different across seasons (Figure S6 in Supporting Information S1) due to the interplay of the two flows. The smallest variations are in winter, which is reasonable considering the low flow of both the Yangtze River and Poyang Lake (Figure S6 in Supporting Information S1).

The above results clearly indicate the backwater effects of the Poyang Lake exerted on the Yangtze River, mainly leading to elevated water levels and associated declined WSS upstream. We can also see that the variation of WSS\_3–5 is also large, indicating a larger spatial extent of the backwater effects. Below, we will further explore how far upstream the backwater affects.





**Figure 3.** Temporal variations of water surface slope (WSS) and the ratio of discharge for 10 reaches under multi-year average conditions (2007–2020). Lines representing WSS refer to the left *y*-axis, and gray shading areas representing Q5/Q1 refer to the right *y*-axis. (a) to (d) shows reaches with upstream gauge station GS\_4, GS\_3, GS\_2, and GS\_1, respectively. Pearson's correlation coefficient and *p*-value between WSS and the Q5/Q1 for each river reach have been labeled in the corresponding legends. The color scheme is consistent with Figure 2a.

#### 3.2. Spatial Extent of Backwater Affected Reaches

As shown in Figure 3, the reaches upstream of the confluence are grouped into four based on the upstream gauge station of each reach. Overall, the WSS of all reaches is well negatively correlated with the flow ratio, indicating a potential backwater effect. As shown in Figure 3a, we know that this reach is clearly affected by backwater, but the correlation between WSS and Q5/Q1 is not perfect (r = -0.53). This can be explained by the unequal impacts of backwater extend on GS\_4 and GS\_5, which leads to different rises in water levels at the gauges. This means that the backwater extends further upstream gauge 4, which can be proved by Figure 3b. Both WSS\_3–5 and WSS\_3-4 decline as the flow ratio increases. One may argue that this might be due to upstream flood events. However, this is not possible in multi-year conditions. Moreover, the difference ( $\Delta$ WSS\_35–34) between WSS\_3–5 and WSS\_3–5 and WSS\_3–4 also declines instead of constant. In the latter case, it will be likely due to floods. The reason for this varying  $\Delta$ WSS\_35–34 is that the water level rising rate is also changing and larger at GS\_5 than at GS\_4. As shown in Figure 3b, in January and February, RR\_3–5 is steeper than RR\_3–4. As the flow ratio increases, RR\_3–5 is getting less steep, and in June, it is almost the same as RR\_3–4, indicating that the backwater is stronger at GS\_5 than GS\_4. Nevertheless, this analysis suggests that the backwater can extend up to GS\_4, although not that strong compared to GS\_5. This is consistent with Wang et al. (2013), that the Poyang Lake outflow caused strong backwater effects on the upstream vicinity of Jiujiang (i.e., GS\_4).

Taking longer reach into consideration, the WSS variations are very different, as shown in Figure 3c. The temporal patterns of WSS\_2–3, WSS\_2–4, and especially WSS\_2–5 are very similar to that of WSS\_3–4 and WSS\_3–5 before mid-May but differ after. The reason is that Q1 increases sharply in May and peaks in late July, while Q5 increases in March and peaks in late June (Figure S5b in Supporting Information S1). Besides, several tributaries recharge the Yangtze River upstream GS\_2. Such configuration leads to complex interplay at reaches



**Figure 4.** The correlation matrix of median water surface slope (WSS) with median Q values under the multi-year (2007–2020) condition. The labels on the top and left indicate the gauge station thus, the element (GS\_2, GS\_1), for example, is the correlation for reach RR\_1–2. The symbol " $\times$ " indicates the correlation is not significant at 0.05 level. The size of the circles indicates the magnitude of the correlation coefficients, as numbered in the elements below the main diagonal. (a) The correlation between Q1 and the WSS, and (b) The correlation between Q5 and the WSS.

RR\_2–5 between upstream floods and downstream backwater. For instance, the rise of WSS\_2–3 after May is much larger than WSS\_2–4 and WSS\_2–5 due to the increased upstream flow, while due to the backwater effects, the latter two are not that much (Figure 3c). Therefore, GS\_2 is likely more affected by upstream flow, while GS\_3 is more affected by backwater, at least in the summer period. Figure 3d shows that backwater at GS\_1 is very likely marginal if not free.

We further explored the potential effects of upstream flow and downstream backwater by analyzing the correlation between flows from the upstream Yangtze River and Poyang Lake and WSS, as shown in Figure 4. In general, the closer to GS\_1 (Q1), the stronger the correlation between Q1 and WSS, indicating larger flow increases the WSS. One exception is WSS\_1–2, which shows a weaker correlation with Q1. This can be attributed to the influences of six major tributaries between GS\_1 and GS\_2 (Figure 1). However, no large lateral inflows downstream GS\_2 recharge into the Yangtze River (see Text S5 and Figure S7 in Supporting Information S1). All reaches starting from GS\_2 are strongly correlated with Q1, showing the dominant control of upstream flow at GS\_2. Moreover, we can see that WSS\_3–5 and WSS\_4–5 are negatively correlated with Q1, indicating that backwater is the dominant factor influencing WSS at these two reaches. This is in line with the above findings.

Figure 4b shows the effect of Poyang Lake flow (Q5) on WSS. There are no statistical correlations between WSS\_1–2 and Q5, and the correlations between WSS\_1–3, WSS1-4, and Q5 are weak, indicating no backwater effect at GS\_1. This is also in line with Figure 3d. The correlations between WSS\_2–3, WSS2-4, and Q5 are positive but weaker than with Q1, suggesting a dominant control of upstream flow at GS\_2. In contrast, both WSS\_3–5 and WSS\_4–5 have strong negative correlations with Q5, suggesting the backwater effect. While WSS\_3–4 negatively correlates with Q5, it very weakly correlates with Q1, indicating that RR\_3–4 is still dominated by downstream backwater.

In short, the backwater can extend up to around GS\_3, which is 74 km away from the confluence. However, the reach RR\_23 may still be affected by the backwater, especially before March (Figure 3c). Because GS\_2 is greatly affected by its upstream inflows (mainstem and tributaries). Therefore, the backwater effect cannot be distinguished.

#### 3.3. Elevated Water Levels Due to Backwater Effects

Due to the lack of river bathymetry, we only focused on the section at GS\_4. As mentioned in the method section, the uncertainty was considered by adopting different Manning's n values. Figure 5 shows the elevated water levels, that is,  $y_{\text{diff}}$ , using different values of Manning's n. Clearly, Manning's n has a large impact on the elevated water levels. Nevertheless, the elevated water level increase with the decrease of WSS in all scenarios (Figure 5).





A: [0.9 - 1.4] B:[1.4 - 1.9] C:[1.9 - 2.4] D:[2.4 - 2.9] E:[2.9 - 3.4] F:[3.4 - 3.9] G:[3.9 - 4.3]

Figure 5. Estimates of raised water level, that is,  $y_{diff}$ , grouped into different water surface slope intervals under 18 sets of minimum and maximum Manning's n values. Symbols A to G represent the seven different WSS\_4–5 intervals. The purple dots indicate the data points of the water depth difference  $y_{diff}$ . Note that here  $S_0 = 4.3$  cm/ km and Q4 from March to October of each year from 2011 to 2020 were used.

And the strong backwater (higher  $y_{diff}$ ) generally leads to a very flat slope, that is, 0.9–1.4 cm/km. The range of the max values of  $y_{diff}$  for different combinations is 2.6–14.1 m (Figure 5). This means that in the worst case,  $y_{diff}$  can be up to 14.1 m (Figure 5a), while in the best case the max  $y_{diff}$  is still up to 2.6 m (Figure 5r). In the most reasonable case, that is, Manning's *n* taking the value between 0.015 and 0.055, the highest elevated water level is about 7.1 m (Figure 5h). One might notice that negative  $y_{diff}$  are shown in some cases. This can be explained by either unreasonable *n* values used or the poor power-law fitting equation for low flow seasons. This can be clearly seen in years 2013 and 2017 (Figures S3 and S8 in Supporting Information S1, refer to Text S6 in Supporting Information S1 for more details).

WSS\_4–5 and the flow ratio show a strong negative correlation in all years except 2020 (Figure S9 in Supporting Information S1), indicating that given similar Yangtze River flow (Q4), a larger flow (Q5) from Poyang Lake will result in stronger backwater effects, leading to a smaller WSS. In other words, if the WSS\_4–5 is continuously monitored, the backwater and elevated water levels can be estimated. Regarding the 2020 case, the Pearson correlation coefficient between WSS\_4–5 and the flow ratio is strong before 17 June 2020. However, due to the quick increase of Yangtze River flow since 17 June 2020, the Poyang Lake flow was reversed, resulting in a negative flow ratio (Figure S10 in Supporting Information S1). The complex interactions of the two flows over the period thereafter leads to a poor overall correlation between WSS\_4–5 and discharge in 2020.

# 4. Discussion and Conclusions

In this study, we revisited the Poyang Lake-Yangtze River interaction from a new perspective. Previous studies have mainly focused on the impact of the Yangtze River on Poyang Lake, while the reverse has been studied less. Using water levels and associated water surface slopes, we first identified the pronounced backwater effects of Poyang Lake outflows on the Yangtze River. Then, we quantified the spatial extent of such effects. We also estimated the water levels raised at the confluence due to such phenomena.

We found that the backwater is pronounced at the confluence of the Poyang Lake-Yangtze River, leading to halved and doubled water surface slopes upstream and downstream the confluence during summer seasons. Further analyses indicate that the backwater can extend much further upstream to around 74 km at the Matouzhen station (GS\_3) and may be even further to Huangshigang (147 km, GS\_2) in March. Nevertheless, we also found that backwater can result in the highest elevated water level up to 7.1 m in the most reasonable case while 14.1 m in the worst case at the Jiujiang station (GS\_4), which is 24 km away from the confluence.

The results indicate that by observing the WSS, say from SWOT (Surface Water and Ocean Topography) mission (Biancamaria et al., 2016) for example, we are able to detect whether backwater event occurs. This might be useful to monitor the interactions between the Yangtze River and Poyang Lake and inform the flood conditions. Moreover, the findings also indicate that pronounced backwater events can occur when the Poyang Lake outflow is large and the ratio between the Yangtze River flow and Poyang Lake outflow is high. Such events often lead to elevated water levels upstream the confluence and consequently, increase the flood risks. To reduce the risk brought by backwater events, we need to reduce either the lake outflow or the ratio of two flows. In this context, lake flow or river flow has to be regulated. For instance, building a dam on the Poyang Lake might solve this issue.

However, due to limited gauges and cross-section data, we are still not able to determine the precise location where the backwater effects extend to. Moreover, the extent may also vary in time due to the combination of the flows of Yangtze River and Poyang Lake. In the future, with detailed river bathymetry and finer-scale longitudinal profiles such as that from SWOT, we can better quantify the extent of backwater effects and the resulting elevated water levels along the river. Similarly, the raised water level of Poyang Lake may also be investigated using SWOT data. Although our study provides an estimation of the backwater effects from a spatial perspective, we are unable to fully capture the temporal dynamics of the backwater effect. In reality, the backwater effect takes time to propagate upstream. This may be addressed by using higher spatiotemporal resolution data in the future.

In a warming climate, extreme flood events will likely become more frequent (J. Chen et al., 2023; Huang et al., 2024), which will likely induce stronger backwater at the confluence of the Poyang Lake and Yangtze River. Therefore, the backwater affected reach will likely be more upstream, and the associated water levels will also be much higher. Such situations will increase flood risks and challenge the current flood-prevention infrastructure of the Yangtze River. Therefore, a better understanding of the river-lake interaction is of great importance. Moreover, in the future, continuous monitoring of river long profile dynamics from space, for example, SWOT would be very helpful.

# **Data Availability Statement**

Data used in this study were mainly downloaded from the Hubei Hydroinfo website (https://hbwater.wetruetech. com/water/portal/water\_details2).

# References

- Abate, M., Nyssen, J., Steenhuis, T. S., Moges, M. M., Tilahun, S. A., Enku, T., & Adgo, E. (2015). Morphological changes of Gumara River channel over 50 years, upper Blue Nile basin, Ethiopia. *Journal of Hydrology*, 525, 152–164. https://doi.org/10.1016/j.jhydrol.2015.03.044 Alves, L. G. S., da Silva, D. D., Vauchel, P., Fraizy, P., & Filizola, N. P. (2020). Variable backwater and channel roughness: Effects on Solimões
- River discharge. Comptes Rendus Geoscience, 352(3), 185–198. https://doi.org/10.5802/crgeos.35
- Biancamaria, S., Lettenmaier, D. P., & Pavelsky, T. M. (2016). The SWOT mission and its capabilities for land hydrology. *Surveys in Geophysics*, 37(2), 307–337. https://doi.org/10.1007/s10712-015-9346-y
- Castelltort, F. X., Bladé, E., Balasch, J. C., & Ribé, M. (2020). The backwater effect as a tool to assess formative long-term flood regimes. *Quaternary International*, 538, 29–43. https://doi.org/10.1016/j.quaint.2020.01.012
- Chen, J., Shi, X., Gu, L., Wu, G., Su, T., Wang, H.-M., et al. (2023). Impacts of climate warming on global floods and their implication to current flood defense standards. *Journal of Hydrology*, 618, 129236. https://doi.org/10.1016/j.jhydrol.2023.129236
  - Chen, M., Zeng, S., Yang, L., Tang, X., & Xia, J. (2022). Detailed attribution of long-term water exchange in the Yangtze-Poyang system at multiple time scales. *Journal of Hydrology*, *612*, 128183. https://doi.org/10.1016/j.jhydrol.2022.128183
  - Chow, V. T. (1959). Open channel hydraulics. McGraw-Hill Book Company, Inc.

Edwards, B. L., Keim, R. F., Johnson, E. L., Hupp, C. R., Marre, S., & King, S. L. (2016). Geomorphic adjustment to hydrologic modifications along a meandering river: Implications for surface flooding on a floodplain. *Geomorphology*, 269, 149–159. https://doi.org/10.1016/j. geomorph.2016.06.037

- Feng, L., Hu, C., Chen, X., & Zhao, X. (2013). Dramatic inundation changes of China's two largest freshwater lakes linked to the three gorges dam. Environmental Science & Technology, 47(17), 9628–9634. https://doi.org/10.1021/es4009618
- Guo, H., Hu, Q., Zhang, Q., & Feng, S. (2012). Effects of the three gorges dam on Yangtze River flow and river interaction with Poyang Lake, China: 2003–2008. Journal of Hydrology, 416–417, 19–27. https://doi.org/10.1016/j.jhydrol.2011.11.027
- Hidayat, H., Vermeulen, B., Sassi, M. G., Torfs, P. J. J. F., & Hoitink, A. J. F. (2011). Discharge estimation in a backwater affected meandering river. *Hydrology and Earth System Sciences*, 15(8), 2717–2728. https://doi.org/10.5194/hess-15-2717-2011

#### Acknowledgments

This work was partially supported by the CRSRI Open Research Program (Program SN: CKWV20221009/KY), Shenzhen Key Laboratory of Precision Measurement and Early Warning Technology for Urban Environmental Health Risks (ZDSYS20220606100604008), High-level University Special Fund (G03050K001), and the SUSTech Research Start-up Grants (Y01296129 and Y01296229). We thank the Editor and reviewers for the insightful comments.

- Hu, Q., Feng, S., Guo, H., Chen, G., & Jiang, T. (2007). Interactions of the Yangtze river flow and hydrologic processes of the Poyang Lake, China. Journal of Hydrology, 347(1–2), 90–100. https://doi.org/10.1016/j.jhydrol.2007.09.005
- Huang, W., Xie, T., Huang, L., Hao, S., & Duan, Y. (2024). Why are extreme precipitation events becoming more frequent in a warming world? *Fundamental Research*. https://doi.org/10.1016/j.fmre.2024.11.009
- Jiang, L., Bandini, F., Smith, O., Klint Jensen, I., & Bauer-Gottwein, P. (2020). The value of distributed high-resolution UAV-borne observations of water surface elevation for river management and hydrodynamic modeling. *Remote Sensing*, 12(7), 1171. https://doi.org/10.3390/ rs12071171
- Lamb, M. P., Nittrouer, J. A., Mohrig, D., & Shaw, J. (2012). Backwater and river plume controls on scour upstream of river mouths: Implications for fluvio-deltaic morphodynamics. *Journal of Geophysical Research*, 117(F1). https://doi.org/10.1029/2011JF002079
- Lesack, L. F. W., & Marsh, P. (2010). River-to-lake connectivities, water renewal, and aquatic habitat diversity in the Mackenzie River Delta. Water Resources Research, 46(12). https://doi.org/10.1029/2010WR009607
- Li, X., Ye, X., Li, Z., & Zhang, D. (2023). Hydrological drought in two largest river-connecting lakes in the middle reaches of the Yangtze River, China. *Hydrology Research*, 54(1), 82–98. https://doi.org/10.2166/nh.2023.110
- Liu, J., Bauer-Gottwein, P., Frias, M. C., Musaeus, A. F., Christoffersen, L., & Jiang, L. (2023). Stage-slope-discharge relationships upstream of river confluences revealed by satellite altimetry. *Geophysical Research Letters*, 50(23). https://doi.org/10.1029/2023GL106394
- Luo, X., Li, H.-Y., Leung, L. R., Tesfa, T. K., Getirana, A., Papa, F., & Hess, L. L. (2017). Modeling surface water dynamics in the Amazon basin using MOSART-inundation v1.0: Impacts of geomorphological parameters and river flow representation. *Geoscientific Model Development*, 10(3), 1233–1259. https://doi.org/10.5194/gmd-10-1233-2017
- Meade, R. H., Rayol, J. M., Conceicão, S. C., & Natividade, J. R. G. (1991). Backwater effects in the Amazon river basin of Brazil. Environmental Geology and Water Sciences, 18(2), 105–114. https://doi.org/10.1007/BF01704664
- Petersen-Øverleir, A., & Reitan, T. (2009). Bayesian analysis of stage-fall-discharge models for gauging stations affected by variable backwater. *Hydrological Processes*, 23(21), 3057–3074. https://doi.org/10.1002/hyp.7417
- Poff, N. L. (2019). A river that flows free connects up in 4D. Nature, 569(7755), 201-202. https://doi.org/10.1038/d41586-019-01393-2
- Sutcliffe, J., & Brown, E. (2018). Water losses from the sudd. *Hydrological Sciences Journal*, 63(4), 527–541. https://doi.org/10.1080/02626667. 2018.1438612
- Tahmid, A., Rahman, M. H., Mounota, S., & Abid Ahsan, K. (2021). Relationship between Manning roughness coefficient and flow depth in Bangladesh rivers. *Malaysian Journal of Civil Engineering*, 33(3). https://doi.org/10.11113/mjce.v33.17363
- Tate, E. L., Sene, K. J., & Sutcliffe, J. V. (2001). A water balance study of the upper White Nile basin flows in the late nineteenth century. *Hydrological Sciences Journal*, 46(2), 301–318. https://doi.org/10.1080/02626660109492823
- Trigg, M. A., Wilson, M. D., Bates, P. D., Horritt, M. S., Alsdorf, D. E., Forsberg, B. R., & Vega, M. C. (2009). Amazon flood wave hydraulics. Journal of Hydrology, 374(1–2), 92–105. https://doi.org/10.1016/j.jhydrol.2009.06.004
- Wang, J., Sheng, Y., Gleason, C. J., & Wada, Y. (2013). Downstream Yangtze River levels impacted by three gorges dam. Environmental Research Letters, 8(4), 044012. https://doi.org/10.1088/1748-9326/8/4/044012
- Wang, J., Sheng, Y., & Wada, Y. (2017). Little impact of the Three Gorges Dam on recent decadal lake decline across China's Yangtze Plain. Water Resources Research, 53(5), 3854–3877. https://doi.org/10.1002/2016WR019817
- Yang, G., Zhang, Q., Wan, R., Lai, X., Jiang, X., Li, L., et al. (2016). Lake hydrology, water quality and ecology impacts of altered river–lake interactions: Advances in research on the middle Yangtze River. *Hydrology Research*, 47(S1), 1–7. https://doi.org/10.2166/nh.2016.003
- Zhang, Q., Xue, C., & Xia, J. (2023). Impacts, contributing factors and countermeasures of extreme droughts in Poyang Lake. *Bulletin of the Chinese Academy of Sciences*, *38*(12). https://doi.org/10.16418/j.issn.1000-3045.20230813005
- Zhang, X., Bi, Z., Sun, X., Wang, P., Xu, Z., & Jia, B. (2023). Backwater effects in rivers and lakes: Case study of dongping lake in China. *Water*, 15(21), 3850. https://doi.org/10.3390/w15213850